Author's response to reviews

Title: Primary pleural leiomyosarcoma with rapid progression and fatal outcomes: a case report

Authors:

ghizlane rais (medghiz@hotmail.com)
soundouss raissouni (soundouss13@hotmail.com)
houda mouzount (hmouzount@hotmail.fr)
meryem aitelhaj (m_meryem_a@hotmail.com)
hind Mrabti (mrabti_h@yahoo.fr)
ahmed jelthi (jelthi@menara.ma)
hassan Errihani (h_errihani@yahoo.fr)

Version: 2 Date: 10 December 2011

Author's response to reviews: see over
Dear Editor

Thank you for your consideration of our manuscript. Please find attached a revised version.

We have formatted the main text to be conform to the journal style.

The patient ethnicity was included

We have formatted the title following the structure: A presenting with B in C: a case report

The header “Case report” was replaced with “Case presentation

We confirm that we have obtained written informed consent from the patient’s next of kin for publication of the case. The consent statement was done accordingly.

We have removed the text from figure two which may reveal the patient’s identity

Thank you for your time and consideration;

I look forward to hearing from you.

Sincerely, yours

Dr Rais Ghizlane